Robot-assisted training compared with an enhanced upper limb therapy programme and with usual care for upper limb functional limitation after stroke: the RATULS three-group RCT

Helen Rodgers,^{1,2,3*†} Helen Bosomworth,^{1*†} Hermano I Krebs,⁴ Frederike van Wijck,⁵ Denise Howel,⁶ Nina Wilson,⁶ Tracy Finch,⁷ Natasha Alvarado,⁸ Laura Ternent,⁶ Cristina Fernandez-Garcia,⁶ Lydia Aird,² Sreeman Andole,⁹ David L Cohen,¹⁰ Jesse Dawson,¹¹ Gary A Ford,^{1,12,13} Richard Francis,¹ Steven Hogg,¹⁴ Niall Hughes,¹⁵ Christopher I Price,^{1,2} Duncan L Turner,¹⁶ Luke Vale,⁶ Scott Wilkes¹⁷ and Lisa Shaw¹

- ¹Stroke Research Group, Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK
- ²Stroke Northumbria, Northumbria Healthcare NHS Foundation Trust, North Tyneside, UK ³Newcastle upon Tyne Hospitals NHS Foundation Trust, Newcastle upon Tyne, UK
- ⁴Mechanical Engineering Department, Massachusetts Institute of Technology, Cambridge, MA. USA
- ⁵School of Health and Life Sciences, Glasgow Caledonian University, Glasgow, UK
- ⁶Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK ⁷Nursing, Midwifery and Health, Northumbria University, Newcastle upon Tyne, UK ⁸School of Healthcare, University of Leeds, Leeds, UK
- ⁹Barking, Havering and Redbridge University Hospitals NHS Trust, Romford, UK ¹⁰London North West University Healthcare NHS Trust, London, UK
- ¹¹Institute of Cardiovascular and Medical Sciences, University of Glasgow, Glasgow, UK ¹²Medical Sciences Division, University of Oxford, Oxford, UK
- ¹³Oxford University Hospitals NHS Foundation Trust, Oxford, UK
- ¹⁴Lay investigator (contact Stroke Research Group, Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK)
- ¹⁵NHS Greater Glasgow and Clyde, Glasgow, UK
- ¹⁶School of Health, Sport and Bioscience, University of East London, London, UK ¹⁷School of Medicine, University of Sunderland, Sunderland, UK

*Corresponding author helen.rodgers@newcastle.ac.uk and helen.bosomworth@newcastle.ac.uk †Joint first authorship

Declared competing interests of authors: Helen Rodgers reports personal fees from Bayer AG (Leverkusen, Germany) outside the submitted work and was a member of the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Clinical Evaluations and Trials (CET) panel (2010–14). Gary A Ford reports personal fees from Amgen Inc. (Thousand Oaks, CA, USA), Bayer AG, Stryker Corporation (Kalamazoo, MI, USA) and Pfizer Inc. (New York, NY, USA), and grants and personal fees from Medtronic plc (Dublin, Ireland), outside the submitted work. Hermano I Krebs reports other financial activity from Interactive Motion Technologies Inc. (Boston, MA, USA) and 4Motion Robotics (Waterton, MA, USA) outside the submitted work. In addition, he has a patent issued (Interactive Robotic Therapist; US Patent 5,466,213; 1995; Massachusetts Institute of Technology) and a patent (Wrist And Upper Extremity Motion; US Patent No. 7,618,381; 2009; Massachusetts Institute of Technology) licensed to Bionik Laboratories Corp. (Toronto, ON, Canada). Denise Howel reports membership of the NIHR Programme Grants for Applied Research Programme subcommittee (2016 to present), and was a member of the NIHR Health Services and Delivery Research Programme Commissioning Board (2012-15). Jesse Dawson reports other financial activity from MicroTransponder Inc. (Austin, TX, USA) outside the submitted work. Luke Vale was a member of the NIHR HTA CET panel (2015-18).

Published October 2020 DOI: 10.3310/hta24540

Plain English summary

The RATULS 3-group RCT Health Technology Assessment 2020; Vol. 24: No. 54 DOI: 10.3310/hta24540

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

M any people who have arm weakness following a stroke feel that insufficient attention is paid by rehabilitation services to recovery of their arm. Unfortunately, it is currently unclear how best to provide rehabilitation to optimise recovery, but robot-assisted training and therapy programmes that focus on practising functional tasks are promising and require further evaluation.

The Robot-Assisted Training for the Upper Limb after Stroke (RATULS) trial evaluated three approaches to rehabilitation for people with moderate or severe difficulty using their arm. These approaches were robot-assisted training using the Massachusetts Institute of Technology-Manus robotic gym system (InMotion commercial version, Interactive Motion Technologies, Inc., Watertown, MA, USA), an enhanced upper limb therapy programme based on repetitive practice of functional tasks and usual care. Robot-assisted training and the enhanced upper limb therapy programme were provided in an outpatient setting for 45 minutes per session, three times per week, for 12 weeks, in addition to usual care.

The Massachusetts Institute of Technology-Manus robotic gym system was selected as it was felt to be the best available technology. The participant sits at a table, places their affected arm onto the Massachusetts Institute of Technology-Manus arm support and attempts to move their arm to play a game on the computer screen. Movements are assisted by the Massachusetts Institute of Technology-Manus if the patient cannot perform the movements themselves.

The results of the RATULS trial show that robot-assisted training did not result in additional improvement in stroke survivors' arm use when compared with the enhanced upper limb therapy programme or usual care. Stroke survivors who received enhanced upper limb therapy experienced meaningful improvements in undertaking activities of daily living, when compared with those participants who received either robotassisted training or usual care. Participants who received enhanced upper limb therapy also experienced benefits in their mobility, compared with usual care participants.

Participants and therapists found both therapies acceptable, and described various benefits. A health economic analysis found that neither robot-assisted training nor the enhanced upper limb therapy programme was a cost-effective treatment for the NHS.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Rodgers *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.370

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 11/26/05. The contractual start date was in January 2014. The draft report began editorial review in December 2019 and was accepted for publication in May 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2020. This work was produced by Rodgers *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

Editor-in-Chief of Health Technology Assessment and NIHR Journals Library

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk